Author’s response to reviews

Title: Women's Approval of Domestic Physical Violence Against Wives: Analysis of the Ghana Demographic and Health Survey

Authors:

David David (dokudavid@gmail.com)

KwakuOppong Asante (kwappong@gmail.com)

Version: 2 Date: 18 Nov 2015

Author’s response to reviews:

Editor's Comments:

1. Title Page: Please include all the co-authors' email addresses in the Title page.

   Response: All co-authors’ email addresses has now been included in the Title Page. See page 1.

2. Tables Reposition. Please reposition the Tables after reference list.

   Response: All the 3 Tables are now repositioned after the reference list. Where they should be placed within text has also been included. See pages, 8 and 9.

3. Please confirm if you obtained relevant permission to use the data and information from the survey and confirm whether these data were publicly available and to add a statement in the manuscript to reflect this.

   Response: We obtained relevant approval to use the data which is publicly available. Additionally, we have acknowledge Measure DHS for access to Ghana Demographic Health Survey’s unrestricted survey data files, which it is authorized to distribute, at no cost, for legitimate academic research.